Title: Behavioral factors to include in guidelines for lifelong oral healthiness: an observational study in Japanese adults

Authors:

Ichizo Morita (i.morita@ucl.ac.uk)
Haruo Nakagaki (nakagaki@dpc.aichi-gakuin.ac.jp)
Atsushi Toyama (at@toyama-dental.com)
Matsumi Hayashi (kobayashi-204@vill.tobishima.lg.jp)
Miho Shimozato (simozato-148@vill.tobishima.lg.jp)
Tsuyoshi Watanabe (nabe55@kvp.biglobe.ne.jp)
Shimpei Tohmatsu (stohmatsu@yahoo.co.jp)
Junko Igo (jiyunko_igo@pref.aichi.lg.jp)
Aubrey Sheiham (a.sheiham@ucl.ac.uk)

Version: 6 Date: 14 December 2006

Author's response to reviews: see over
Dr Jo Appleford  
Senior Assistant Editor  
BMC Oral Health

14th December, 2006

RE: MS:  6207188421074429
Behavioral factors to include in guidelines for lifelong oral healthiness: an observational study in Japanese adults

Dear Dr Appleford,
Thank you very much for your letter of December 8th, 2006 saying that you have accepted our paper. I have uploaded a final version of the above mentioned paper that has some minor corrections.

Yours sincerely,

Dr Ichizo Morita

Department of Epidemiology and Public Health,  
University College London  
1-19 Torrington Place, London WC1E 6BT  
UK  
Tel 44-20-7679-1253  
Fax 44-20-7813-0280  
e-mail: i.morita@ucl.ac.uk